\begin{table}%t1
\caption{\label{tab1}Fe II radiative lifetimes.}
%\begin{center}
\par

\resizebox{14.5cm}{!}{\includegraphics[clip]{irontau2.eps}}

%\end{center}

{[1]~\citet{BBKAP91}, $~~$ [2]~\citet{GAPJB92}, $~~$ [3]~\citet{HLGN92},
$~~$ [4]~\citet{SMH88},}\\ {$~~$  [5]~\citet{LLSJ99}.\\
* Lifetime was measured with SBS--compressed laser pulses.}
\end{table}
